Notes of the RICH1 Phone meeting 26/10/2007

People Present:

CERN: Carmelo, Olav, Jon, Fabio. 

IC: Bill, Dave, Richard.
Oxford: Neville

Apologies for absence:

Werner

Discussed Agenda Topics

1. Gas Enclosure

The upper HPD rail system was successfully re-installed last week.  Now awaiting the arrival of the HPD module in 2-3 weeks, with the test of the whole unit to happen after the VELO installation has been completed.

2. Mirrors

Testing of the spherical mirror sample has been postponed until mid-November due to people being away from CERN.  The amount of C4F10 used in the storage of the mirror sample has been estimated at approx. 10-15 kg of gas per year.  

3. HPD assembly

3.1 Upper HPD box

Trevor is producing drawings and components are currently being machined at Imperial College with approx. 2 weeks until final assembly.  Testing of the upper rail system was successful, but the some of the modifications made will need to wait until arrival of the HPD box until they can be checked.

3.2 Lower HPD rail system

The positions of the rail coordinates have been agreed upon.  Material has arrived in Oxford, and construction of the frame has started.  Brian is retiring at the end of the month and the drawings will be finished by Joe at Oxford.

 3.3 HPD cabling

LV cables were moved slightly to allow easier access for the VELO installation.  No more cabling to be done until the HPD module has arrived. Upper fibres and individual LAMS camera cables still to be installed.  A test of the specs chain is being organized in the lab, which will simulate the situation in the pit.  The first of the light detector phototubes has arrived and has been powered up in the lab with good results.  The 2nd tube is due in the UK on the 1st November.

3.4 Mechanics

The majority of the HPD mechanics have been anodized and assembly is in progress.

4.0 LAMS

A rough design of the mounting brackets has been completed.  Drawings will be provided later.  The mountings will be designed in such a way that they can be removed for extraction of the mirrors and aerogel.
5.0 MDMS

Good progress from Ray on the design of the fixing into the photon funnel. The design of the photon funnel will need to be changed in light of recent measurements to fit between the HPD module, gas enclosure and the shielding
9. Next meeting

Friday 9th November 2007 at 10:00(GMT) 11:00(CET).

